

October 28, 2013

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 5588-review.doc).

Title: FDG uptake in absence of CT abnormality on PET-CT: what is it?

Author: Yiyan Liu

Name of Journal: World Journal of Radiology

ESPS Manuscript No: 5588

The manuscript has been improved according to the suggestions of the reviewers:

1. Format has been updated.
2. A typing error was corrected on the page 7 (on previous submission, page 8).
3. The paragraph about osseous uptake is included in the section "Real lesion not seen on CT".

Thank you again for publishing my manuscript in World Journal of Radiology.

Sincerely yours,

Yiyan Liu, MD, PhD
Associate Professor
Department of Radiology
New Jersey Medical School
Rutgers University
150 Bergen Street, H141
Newark, NJ 07101
Unites States
1-973-972-6022 (phone)
1-973-972-6954 (fax)
E-mail: liuyl@njms.rutgers.edu